Author’s response to reviews

Title: New PRSS1 and common CFTR mutations in a child with acute recurrent pancreatitis, could be considered an "Hereditary" form of pancreatitis?

Authors:

Vito D Corleto (vito.corleto@uniroma1.it)
Stefano Gambardella (stefanogambardella@gmail.com)
Francesca Gullotta (francesca.gullotta@hotmail.com)
Maria R D'Apice (lucidi@opbg.net)
Matteo Picciucchi (matteopiciucchi@libero.it)
Elena Galli (galli.elena@fbfrm.it)
Vincenzina Lucidi (lucidi@opbg.net)
Giuseppe Novelli (novelli@med.uniroma2.it)
Gianfranco Delle Fave (gianfranco.dellefave@uniroma1.it)

Version: 7 Date: 18 September 2010

Author’s response to reviews: see over
17/09/2010

Dear Editor

The manuscript has been revised as requested.

Best wishes

Vito D. Corleto M. D.